

Format for ANSWERING REVIEWERS



World Journal of Dermatology

March 3, 2014

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: ESPS Manuscript NO: 8283 -review.doc).

Title: Phytotherapy and psoriasis: Complementary and alternative medications

Author: Sami A. Gabr, Ahmad H. Alghadir

Name of Journal: *World Journal of Dermatology*

ESPS Manuscript NO: 8283

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewers

Reviewr No. 00646535

(1) Mention of liver function to be monitored using Chinese Herbal Medicine because of the toxic effect

(2) Chapter 3, Chapter4, and Chapter 5 could be resumed.

(3) The histologic images could be resumed.

Reviewr No. 00646501:

(4) Reducing section 3 on clinico-pathological features which any clinician should know about

(5) Combining section 4 and 5 (PASI is a well known outcome measure).

(6) Deleting figure 1 or 2 dealing with the same issue - deleting table 1 and 2

3 References and typesetting were corrected

Thank you again for publishing our manuscript in *World Journal of Dermatology*

Sincerely yours,

Dr. Sami A. Gabr

Rehabilitation Research Chair (RRC), King Saud University, Riyadh, KSA

Tel.: +966562060018; Fax: +96614698541

E-mail address: nadalab2009@hotmail.com, Dr.Gabr@hotmail.com, sgabr@ksu.edu.sa